Author’s response to reviews
Title: Robotic versus Laparoscopic Gastrectomy for Gastric Cancer: A Systematic Review and Updated Meta-analysis

Authors:
Ke Chen (chenke0301243@163.com)
Yu Pan (panyu1013@126.com)
Bin Zhang (doctorbinzhang@126.com)
Hendi Maher (maherhend@qq.com)
Xian-fa Wang (srrshwxf@163.com)
xiujun cai (caixiujunzju@163.com)

Version: 3 Date: 10 Aug 2017

Author’s response to reviews:
1. Please have the text edited by a professional language editing service or a native English speaking colleague. There are many issues with grammar, wording, spelling, and/or punctuation that need to be addressed. For example, “Our researches has the following limitationss”
Answer: Answer: The revised paper has been polished by a native speaker. Hendi Maher, born in Australia, Master Degree Student of Zhejiang University School of Medicine, 866 Yuhangtang Road, Hangzhou 310058, Zhejiang Province, China. and general surgery department sir run run shaw hospital.

2. Please change your Introduction heading to Background.
Answer: We have amended the paper.

3. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.
Answer: We have amended the paper.